Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
March 17, 2010
Injector 
· Energy 56.62 MeV

· Injector may need optimization prior to delivering high current to Hall A.

· Energy 56.25 MeV

· Injector has been set up (100uA to inline dump)
.
Accelerator   Energy 503.25 MeV
· (3/15) The accelerator energy was increased by 0.65%. The energy change went well. Orbits, path length optics, and pass 1 and pass 3 separations were optimized. 
· 5 pass witness beam from Hall C laser runs to the BSY dump.

· A radiation survey showed that there was a hot spot at 7R05. It was most likely due to bad optics (not orbit). We took care of it during yesterday’s energy change.

· Over the weekend there was a cooling water leak in the east arc in the tunnel. It was addressed Monday (3/15)

· There are vacuum spikes on the vacuum commander screen. The spikes were there before the restoration of beam. 
· 5 pass tuned beam to the BSY dump was recovered. Orbits, path length, dispersion and betatron match need more work.
· There were some difficulties during the beam recovery because some cavities were out of phase too much and MQB2L24 polarities were wrong. 
Hall A---Pass 1, Prex from March 11 to May 24, 2010
· The hall will be ready Thursday swing shift (3/18)
· Hall A Moller procedure is being revised
· Hall A may not be ready before March 15.

· Hot Checkout is not done yet

Hall B ---Pass 3, g9-Frost from March 11 to July 22, 2010
· Hall B beam is fine

· MBC2C22H, MBC2C23V, MQA2C22, MQA2C23 and MQA2C24 will be not used in the experiment. They have been degaussed and will be taken out of the control screens.

· Hall B uses photon beam for this experiment meaning that the electron beam goes to its Tagger dump.

· Hall B is on schedule.

· Hot Checkout is not done yet

Hall C---QWeak from May 25 to July 22, 2010
· QWeak installation
